Author's response to reviews

Title: Health Habits and Other Characteristics of Dietary Supplement Users: A Review

Authors:

Annette Dickinson (anneditedickinson@comcast.net)
Douglas MacKay (dmmackay@crnusa.org)

Version: 4 Date: 4 February 2014

Author's response to reviews: see over
February 4, 2014

TO: Editors, *Nutrition Journal*

FROM: Annette Dickinson, Ph.D.

SUBJECT: Manuscript on Health Habits of Dietary Supplement Users

My co-author Douglas MacKay and I are delighted to learn of your acceptance of our revised manuscript of a review article on the health habits of dietary supplement users. Per your request of February 2, 2014, we have further revised the manuscript to include an Acknowledgements section on page 20, following the sections on Competing Interests and Authors’ Contributions.

Thank you for your consideration of our work.